#### Check for updates

#### **OPEN ACCESS**

EDITED BY Nico Sollmann, Ulm University Medical Center, Germany

REVIEWED BY Karupppasamy Subburaj, Aarhus University, Denmark Daniel Strack, Aarhus University, Denmark

\*CORRESPONDENCE Yunhua Luo Yunhua.Luo@umanitoba.ca

RECEIVED 02 December 2024 ACCEPTED 27 January 2025 PUBLISHED 04 March 2025

#### CITATION

Luo Y (2025) Biomechanical perspectives on image-based hip fracture risk assessment: advances and challenges. *Front. Endocrinol.* 16:1538460. doi: 10.3389/fendo.2025.1538460

#### COPYRIGHT

© 2025 Luo. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

# Biomechanical perspectives on image-based hip fracture risk assessment: advances and challenges

### Yunhua Luo<sup>1,2\*</sup>

<sup>1</sup>Department of Mechanical Engineering, University of Manitoba, Winnipeg, MB, Canada, <sup>2</sup>Department of Biomedical Engineering (Graduate Program), University of Manitoba, Winnipeg, MB, Canada

Hip fractures pose a significant health challenge, particularly in aging populations, leading to substantial morbidity and economic burden. Most hip fractures result from a combination of osteoporosis and falls. Accurate assessment of hip fracture risk is essential for identifying high-risk individuals and implementing effective preventive strategies. Current clinical tools, such as the Fracture Risk Assessment Tool (FRAX), primarily rely on statistical models of clinical risk factors derived from large population studies. However, these tools often lack specificity in capturing the individual biomechanical factors that directly influence fracture susceptibility. Consequently, image-based biomechanical approaches, primarily leveraging dual-energy X-ray absorptiometry (DXA) and quantitative computed tomography (QCT), have garnered attention for their potential to provide a more precise evaluation of bone strength and the impact forces involved in falls, thereby enhancing risk prediction accuracy. Biomechanical approaches rely on two fundamental components: assessing bone strength and predicting fall-induced impact forces. While significant advancements have been made in image-based finite element (FE) modeling for bone strength analysis and dynamic simulations of fallinduced impact forces, substantial challenges remain. In this review, we examine recent progress in these areas and highlight the key challenges that must be addressed to advance the field and improve fracture risk prediction.

KEYWORDS

hip fracture, risk assessment, DXA, QCT, bone strength, fall-induced impact force

### 1 Introduction

Hip fractures are a significant health concern, particularly among older adults, who often have a high prevalence of osteoporosis, contributing to substantial morbidity, mortality, and healthcare costs worldwide (1-3). In 2019, there were 178 million new fractures globally, marking a 33.4% increase since 1990, partly driven by population aging (2, 3). Hip fractures constituted a significant proportion of these cases. Projections indicate that the number of hip fractures will nearly double by 2050, underscoring the urgency for

effective fracture risk assessment to identify high-risk individuals and implement preventive measures (4, 5). Accurate assessment of hip fracture risk is crucial, as it enables targeted interventions and support, thereby reducing the burden of these fractures (6).

The current clinical approach for diagnosing hip fracture risk and treating pre-fracture conditions relies primarily on risk factors such as bone mineral density (BMD) and population-based statistical models (7). Although low BMD is widely regarded as a key biomarker for bone fractures, this approach has significant limitations. Studies indicate that approximately 50% of fractures occur in individuals with BMD values above the established threshold (8, 9). BMD also serves as the primary target for many treatment options, particularly those aimed at osteoporosis (10, 11). FRAX is one of the most widely used tools globally to estimate the 10-year probability of hip fractures and other major osteoporotic fractures (12-15). It incorporates several key risk factors, including age, gender, BMD at the femoral neck, prior fractures, parental history of hip fractures, smoking status, alcohol consumption, glucocorticoid use, and rheumatoid arthritis. The predictive accuracy of FRAX has been reported as moderate (16, 17), with area under the receiver operating characteristic (ROC) curve (AUC) values ranging from 0.70 to 0.75 for hip fracture prediction. The tool tends to underestimate fracture risk in certain populations, such as those with frequent falls or advanced age, where fall risk is not fully incorporated (14, 18, 19). The primary limitation of the current tools lies in their reliance on statistical modeling of risk factors. These tools predict fracture risk by identifying broad population-level trends and applying them to individual cases (20).

To improve the accuracy of hip fracture risk assessments, there is a pressing need to develop biomechanical models (21). Image-based biomechanical approaches are theoretically more reliable and accurate than statistical models derived from clinical risk factors because they directly assess the mechanical properties of bone and the forces contributing to fractures (22, 23). Unlike statistical models, which rely on population-level data and indirect associations, biomechanical approaches evaluate individual-specific factors such as bone strength, geometry, and microstructure. These methods utilize advanced imaging techniques, such as high-resolution CT and finite element (FE) modeling, to simulate the mechanical response of bones to applied forces, providing a direct measurement of fracture risk. Furthermore, image-based dynamic simulations can model fallinduced impact forces by analyzing body kinematics, fall trajectories, and surface interactions (24, 25). These simulations allow for a detailed assessment of the magnitude, direction, and distribution of impact forces during a fall (25), which are critical in determining fracture risk. By integrating subject-specific bone properties with dynamic fall scenarios (26), biomechanical approaches can provide a comprehensive and personalized evaluation of fracture risk, addressing limitations in clinical tools that overlook the interplay between bone strength and fall mechanics. This capability highlights their potential to significantly enhance fracture risk assessment and prevention strategies.

Substantial progress has been made in developing image-based biomechanical models for predicting hip fracture risk. However, significant challenges remain, which must be addressed before these biomechanical models can be integrated into clinical practice. This review extensively examines recent advancements and discusses the key challenges that need resolution. The layout of the remainder of this paper is as follows: Section 2 outlines the framework of imagebased biomechanical approaches; Section 3 reviews the progress and challenges in image-based finite element modeling of bone strength; Section 4 explores the advancements and remaining obstacles in image-based dynamic simulation of falls; and Section 5 concludes the review with proposals for future research directions.

# 2 Image-based biomechanical approach to assess hip fracture risk

Based on engineering material mechanics, hip fracture is determined by two key variables (Figure 1): femoral strength and the force applied to the hip, both of which are subject-specific. Femoral strength refers to the maximum force the femur can withstand before fracturing and is primarily determined by the bone's material composition-such as inorganic minerals, organic proteins, and water-along with its macroscopic geometry and microstructural integrity. Since the majority of hip fractures result from falls (27-30), the impact force generated during a fall from standing height is considered in assessing hip fracture risk. This force is influenced by variables such as body height, body mass, and fall orientation and can vary significantly depending on the dynamics of the fall and the compliance properties of the impacted surface. When the fall-induced force exceeds femoral strength, a hip fracture occurs. Accurately determining femoral strength, fall-induced impact force, and their interplay is essential



for developing precise and predictive models of fracture risk, enabling more effective prevention and individualized treatment strategies.

Given the necessity for non-invasive approaches in assessing hip fracture risk, determining both femoral strength and fallinduced impact force must be conducted safely and without invasive procedures. Medical imaging offers an essential solution to this challenge, as illustrated in Figure 1. Advanced imaging technologies, such as dual-energy X-ray absorptiometry (DXA) and quantitative computed tomography (QCT), enable subjectspecific assessment of bone structure, geometry, and tissue composition within the body. These imaging modalities provide critical data on bone mineral density, and material composition, which are essential for estimating femoral strength. Information about trabecular architecture can be partially inferred from QCT data, as it provides 3D volumetric imaging capable of analyzing parameters such as trabecular thickness, separation, and number. However, DXA, being a 2D imaging modality, lacks the resolution to capture detailed trabecular architecture. For more precise insights into individual trabecular microstructure, higherresolution imaging modalities such as micro-CT or HR-pQCT are required, although these are typically limited to in vitro studies or extremities in vivo. Furthermore, imaging can capture patientspecific anatomical and kinetic properties, which can then be used in dynamic simulations to predict fall-induced impact forces. The integration of imaging data into biomechanical models ensures a personalized and accurate evaluation of fracture risk.

Significant advances have been made in the development of image-based finite element (FE) modeling for predicting femoral strength and dynamics simulations for analyzing fall-induced impact forces. While these advancements offer promising opportunities to assess subject-specific fracture risk more accurately, challenges and obstacles remain. The following sections provide a detailed review of these advancements, highlighting the progress achieved and the critical barriers that must be addressed to facilitate their integration into clinical practice.

### 3 Image-based finite element modeling of bone strength

To construct a finite element (FE) model of the femur for determining its strength, several key pieces of information are required. First, accurate geometry of the femur is essential, typically derived from high-resolution medical imaging modalities such as computed tomography (CT). These images provide detailed spatial data that allow for the reconstruction of the femur's shape and structural features, including cortical thickness, trabecular architecture, and overall bone dimensions. Second, the material properties of the bone must be specified, including the elastic modulus, yield strength, and density of both cortical and trabecular bone. These properties are often determined from CTderived Hounsfield units, which can be mapped to bone density and subsequently used to estimate the material properties. Additionally, boundary conditions and loading scenarios must be defined to replicate physiological or fall-related forces acting on the femur, such as compressive loads during standing or oblique forces during a fall. Together, these inputs enable the FE model to simulate stress and strain distributions within the femur and predict its failure point under applied loads.

While DXA and QCT are the primary imaging modalities discussed in this paper due to their clinical relevance for biomechanical modeling, other advanced imaging technologies also hold promise. High-resolution peripheral QCT (HR-pQCT) offers detailed insights into bone microarchitecture but is limited to extremities due to its field of view. Dual-energy CT (DECT) enables improved material characterization by distinguishing between bone mineral density and other components, such as collagen and water. Magnetic resonance imaging (MRI) can provide complementary information on bone marrow composition and trabecular structure but lacks the spatial resolution necessary for finite element modeling of bone strength. Although these techniques have significant potential, their high cost, limited availability, and practical constraints currently limit their widespread application in hip fracture risk assessment.

Numerous finite element (FE) models have been developed for the femur, with most falling into two primary categories: those based on dual-energy X-ray absorptiometry (DXA) and those based on quantitative computed tomography (QCT).

### 3.1 DXA-based finite element models

DXA-based FE models are particularly attractive due to the merits of DXA over QCT, including lower cost, wider availability, and reduced radiation exposure. These models leverage twodimensional (2D) DXA images to estimate femoral strength and fracture risk by incorporating simplified assumptions about bone geometry and material properties, as illustrated in Figure 2. First, a plane stress model (31) or engineering beam model (32) is adopted, representing the femur by projecting all the bone material along the DXA scanning direction, thereby reducing the complex 3D geometry of the femur to a simplified 2D model with uniform thickness. Second, the areal bone mineral density (aBMD) derived from DXA is correlated with key material properties (33), such as bone elasticity and yield stress, enabling the estimation of bone strength in the medial-lateral plane.

In DXA-based finite element analyses, material models primarily assume linear elastic behavior due to the simplicity and computational efficiency required for clinical applicability (36, 37). For instance, the Young's modulus is often estimated based on empirical relationships with areal bone mineral density (aBMD) (36, 37). Some studies incorporate piecewise linear models to account for yield points and post-yield behavior, though these are less common due to the limitations of 2D projections in capturing detailed material heterogeneity. Non-linear models, which consider failure criteria or plasticity, have been less frequently applied in DXA-based FE analyses due to the challenges in accurately representing complex bone material using 2D data (38). DXAbased finite element studies commonly employ simplified yet



clinically relevant loading and boundary conditions to simulate scenarios associated with hip fractures. The most frequently used loading scenario involves a sideways fall, which reflects the most common fall mechanism leading to hip fractures in older adults. In these simulations, the femoral head is typically subjected to oblique forces, representing the impact of the greater trochanter against the ground during a fall. These assumptions and correlations strike a practical balance between prediction accuracy and simplicity, positioning DXA-based FE models as a promising patient-specific tool for assessing hip fracture risk in clinical settings.

The development and improvement of DXA-based finite element models for hip fracture risk assessment have significantly advanced, driven by the need for more accurate and individualized predictions of bone strength and fracture risk. Early studies demonstrated the feasibility of integrating finite element analysis (FEA) with DXA imaging to estimate femoral strength (31, 34, 39). Luo et al. (35) investigated the precision of DXA-based finite element models, identifying body positioning during DXA scanning as a critical factor influencing model accuracy. Further advancements focused on automation and clinical applicability, as illustrated by Luo et al. (40) and Yang et al. (41), who developed fully automated DXA-based FEA tools that not only stratified fracture risk more effectively than femoral neck bone mineral density (BMD) but also streamlined workflows for routine clinical use. Validation efforts, such as those by Dall'Ara et al. (42), confirmed the accuracy of DXA-based FEA models against experimental data, reinforcing their reliability. Simplified 2D FEA models derived from DXA images were also validated against more complex 3D models by Terzini et al. (38), highlighting their practicality with reasonable predictive accuracy. These continuous improvements have established DXA-based FEA as a robust and clinically viable approach to addressing the limitations of traditional BMD-focused fracture risk assessments. DXA-based finite element models are increasingly being utilized for hip fracture risk assessment. Yang et al. (43) demonstrated the effectiveness of this approach in the Osteoporotic Fractures in Men (MrOS) study, where femoral strength estimates derived from FEA showed a strong association with incident fractures. Sarvi and Luo (44) investigated sex differences in hip fracture risk using biomechanical modeling and identified significant distinctions that traditional BMD measurements failed to capture. Additionally, Ferdous et al. (31) underscored the value of patientspecific FEA models in evaluating individualized fracture risk, further highlighting the adaptability and clinical potential of this technique. In addition to risk assessment, DXA-based FE models have been used to monitor the effectiveness of osteoporosis treatments. Mochizuki et al. (45) employed DXA-based hip structural analysis to evaluate changes in bone strength during teriparatide treatment, demonstrating significant improvements in femoral strength over 24 months.

Despite their advantages, DXA-based FE models have several limitations. DXA only provides 2D projections of the femur, which limits the model's ability to capture the 3D geometry and microstructural details essential for accurate stress and strain predictions. DXA-based FE models often rely on oversimplified assumptions about the relationship between aBMD and bone material properties. These assumptions may overlook variations in the spatial distribution of bone mass, including differences in cortical and cancellous component densities, which are critical for capturing the anisotropic nature of femoral strength. Additionally, the 2D nature of DXA imaging restricts its capacity to evaluate trabecular architecture and cortical porosity, both of which are essential determinants of bone strength and fracture risk. Anatomical geometry reconstruction from DXA images models the entire femur as a single entity (46, 47), assigning subject-specific material properties based on areal bone mineral density (aBMD) values derived from DXA images. This method simplifies the geometry and computational requirements but inherently lacks the ability to distinguish between cortical and trabecular compartments, which are critical for accurately capturing the heterogeneity of bone properties. DXA-based FE models often use simplified loading scenarios to estimate femoral strength, which may not accurately represent the complex, multidirectional forces experienced during real-world falls. DXA-based FE models primarily reflect changes in BMD, making them less sensitive to other critical factors, such as improvements in bone collagen quality and the integrity of collagen crosslinks (48, 49), which play a significant role in bone strength and may result from osteoporosis treatments. Variations in DXA scanner calibration and software algorithms (50) can introduce inconsistencies in BMD measurements, affecting the reproducibility of FE model predictions.

### 3.2 QCT-based finite element models

QCT-based finite element models are constructed from threedimensional data acquired through quantitative computed tomography (QCT). While the process of creating QCT-based finite element models shares similarities with that of DXA-based models, as illustrated in Figure 3, the key differences lie in the threedimensional representation of femur geometry and the use of volumetric bone mineral density (vBMD) instead of areal BMD (51). The process begins with acquiring high-resolution QCT images of the femur. These datasets are segmented to differentiate bone tissue from surrounding structures, enabling the extraction of cortical and trabecular bone regions (52). The image data are then converted into 3D finite element meshes, typically composed of tetrahedral or hexahedral elements, to accurately approximate the femoral geometry (53). Tetrahedral elements are more versatile in conforming to complex geometries, making them suitable for irregular structures like the femur. In contrast, hexahedral elements offer higher accuracy and computational efficiency for simpler, structured geometries but are less adaptable to irregular shapes. The choice between the two depends on the trade-off between geometric fidelity and computational efficiency in the modeling process. Bone densities are obtained from QCT image intensities through calibration with phantoms, which provide reference values for converting Hounsfield units into equivalent bone density measures. Material properties are assigned based on the density values using empirical relationships that link density to Young's modulus and other mechanical parameters (51, 54). Boundary and loading conditions are applied to simulate physiological or traumatic scenarios, such as normal gait or sideways falls (52). Finally, these models are solved using numerical methods to estimate stress, strain, and overall femoral strength (53). Overall femoral strength is typically defined as the maximum load the femur can withstand before failure, as determined by the finite element simulation. This definition depends on the material model used; for linear elastic models, it is based on yield stress, while for non-linear models, it may incorporate ultimate stress or fracture criteria. The choice of strength definition varies depending on the specific study objectives and modeling assumptions.



QCT-based finite element models have emerged as a robust tool for studying femoral strength and hip fracture risk. By integrating 3D imaging with advanced computational modeling, these models provide detailed insights into the mechanical behavior of the femur under different loading conditions. Below, we review key research applications of QCT-based FE models in this domain. QCT-based FE models have been widely used to evaluate the structural integrity of the femur under simulated loading conditions. Dragomir-Daescu et al. (56) developed robust models to predict femoral stiffness and fracture load during a sideways fall, demonstrating strong correlations with in vitro experimental data. Mirzaei et al. (57) applied QCT-based FE analysis to analyze strength and failure patterns in the human proximal femur, revealing critical mechanical insights that aid in fracture prediction. Dall'Ara et al. (58) validated nonlinear QCT-based FE models using in vitro human femur, showing their reliability across multiple experimental configurations. These studies highlight the utility of QCT-based FE models in quantifying femoral strength and identifying high-risk individuals.

QCT-based FE modeling has also been employed to study variations in femoral strength across different populations. Shen et al. (59) investigated the relationship between body mass index (BMI) and QCT-derived hip strength in older men, providing biomechanical explanations for the effects of BMI on fracture risk. Black et al. (60) conducted a large prospective study to assess the relationship between proximal femoral structure, as derived from QCT, and hip fracture risk in men, establishing the clinical relevance of QCT-based measurements. Faisal and Luo (55) examined differences in fracture risk between left and right femora using QCT-based FE models, identifying asymmetries that may inform individualized treatment strategies. Several studies have focused on evaluating hip fracture risk under specific mechanical or pathological conditions using QCT-based FE models. For example, Kheirollahi and Luo (61) used cross-sectional strain energy derived from QCT-based FE models to assess hip fracture risk, demonstrating the sensitivity of this method to variations in bone density and geometry. Carpenter et al. (62) emphasized the importance of fall orientation on femoral neck strength, showing that certain fall directions substantially increase fracture risk. Such studies underline the versatility of QCT-based FE models in replicating realistic fracture scenarios.

Traditional metrics for monitoring treatment effects typically include changes in areal bone mineral density (aBMD) as measured by DXA. Treatments such as bisphosphonates or anabolic agents like teriparatide are commonly assessed using these metrics. These methods focus on improving bone density and strength over time, offering a baseline for evaluating therapeutic outcomes. QCT-based FE models are employed to examine the contributions of cortical and trabecular compartments to overall femoral strength. Christiansen et al. (63) used these models to explore age-related changes in bone strength, showing how the cortical and trabecular components contribute differently to mechanical stability in men and women. These findings have enhanced the understanding of how age and sex influence fracture risk. QCT-based FE models have been applied to evaluate the impact of osteoporosis treatments and other clinical conditions on femoral strength. Engelke et al. (64) used these models to monitor regional changes in bone mineral density after ibandronate treatment, demonstrating how such treatments improve hip strength. Similarly, Black et al. (60) showed how QCT-based parameters could predict treatment outcomes more effectively than traditional metrics, emphasizing the potential of these models in clinical decision-making.

QCT-based finite element models provide a more detailed and robust approach than DXA-based models for assessing femoral strength and hip fracture risk. QCT offers greater detail compared to DXA by providing 3D volumetric imaging, allowing separate analysis of cortical and trabecular compartments. Additionally, QCT measures volumetric bone mineral density (vBMD), which is not influenced by bone size or projection errors, and enables assessment of bone geometry, microarchitecture, and material properties with higher spatial resolution. However, QCT-based models are not without limitations. A major challenge lies in the high radiation dose associated with QCT imaging, which restricts its routine clinical use, especially for longitudinal studies (65). Additionally, constructing and solving QCT-based finite element models require advanced computational resources and expertise, which can be a barrier to widespread adoption in clinical practice (56). Variability in imaging protocols and finite element modeling assumptions, such as mesh density and material property assignment, can introduce inconsistencies and limit reproducibility across studies (58, 63). Furthermore, the use of density-based material property assignment often oversimplifies bone's heterogeneous and anisotropic mechanical behavior, potentially reducing the accuracy of predictions (57). Finally, these models generally do not account for dynamic biological processes, such as bone remodeling or microdamage accumulation, which are critical for understanding changes in bone strength over time (61). Addressing these limitations through advancements in imaging, modeling, and computational techniques is essential to enhance the clinical utility of QCT-based finite element models.

# 3.3 Challenges in image-based finite element modeling of femoral strength

Finite element modeling of femoral strength based on medical imaging, such as QCT or DXA, has advanced significantly in recent years, offering valuable insights into bone mechanics and fracture risk. However, despite these advancements, and alongside the limitations discussed in the previous subsections, several critical challenges persist, hindering the accuracy, reliability, and clinical utility of these models. One major issue lies in the challenge of accurately characterizing bone material properties, such as Young's modulus, yield stress, and toughness, from medical images. Another significant challenge is capturing the anisotropic behavior of femoral strength, which varies with loading orientation and is influenced by the direction of impact forces during a fall. Bone anisotropy has been studied in vertebral bones (66), where transverse isotropy is modeled by scaling Young's modulus according to directional properties. Application of a similar approach to the femur requires experimentally derived scaling factors. Addressing these complexities requires advancements in imaging technologies, image-based material characterization algorithms, and modeling techniques, as these elements are pivotal for enhancing the predictive accuracy and reliability of image-based finite element models.

# 3.3.1 Image-based characterization of bone material properties

Accurately characterizing bone material properties, such as Young's modulus, yield stress, and toughness, from DXA or QCT images remains a significant challenge due to the composite nature of bone. Bone is a hierarchical material composed of inorganic minerals (primarily hydroxyapatite), organic proteins (mostly collagen), and water. Each of these components contributes distinct mechanical properties to bone, and their interplay determines the overall strength and toughness of the tissue. However, medical imaging modalities like DXA and QCT are limited in their ability to quantify or assess the quality of these individual components, which hinders precise material characterization.

DXA and QCT provide information about bone density, which is a proxy for the amount of mineral content in bone. However, this metric alone does not capture variations in the organic matrix or water content, both of which critically influence mechanical properties. Studies have shown that the organic matrix, particularly collagen cross-linking, plays a pivotal role in bone toughness and resistance to fracture (67-69). Similarly, bound and free water in bone contribute to its viscoelastic and fatigueresistant properties (70). Limited by their working principles, both DXA and QCT can measure only mineral density, while the characterization of organic proteins and water remains challenging with these imaging modalities. As a result, the contributions of organic proteins and water to bone strength, particularly toughness, are not accounted for in DXA- and QCTbased models (71, 72). Further complicating the issue is the heterogeneity of bone mineralization. The degree of mineralization varies across individuals and regions within the bone, affecting stiffness and brittleness. QCT-based finite element models often rely on empirical density-elasticity relationships derived from bone properties, which may not account for interindividual variability in the inorganic-organic composition or regional differences within the same bone (54, 73). This limitation undermines the ability to predict mechanical properties accurately under diverse physiological or pathological conditions.

Another critical challenge lies in accurately determining the stress-strain curves for the individual components of bone, particularly minerals and proteins, which are dependent on the sub-compositions and sub-microstructure in the components. These curves are fundamental for understanding bone behavior under impact forces but are highly subject-dependent, adding complexity to their precise characterization. For instance, the mechanical behavior of hydroxyapatite, the primary mineral in bone, depends on its crystal size, orientation, and substitutional chemistry, all of which can vary significantly among individuals (74). Similarly, the organic matrix, predominantly composed of type I collagen, shows variability in structure and cross-linking patterns among individuals, directly influencing its mechanical response under load (75). Factors such as age, sex, ethnicity, and health status further modulate the quality and quantity of these bone components, resulting in significant differences in their mechanical properties (67). For example, aging reduces collagen quality while increasing mineral crystallinity, leading to stiffer but more brittle bones (76). Imaging modalities like QCT and DXA currently lack the capability to capture these subtle yet critical changes in bone composition and quality. Furthermore, the absence of standardized methods for characterizing these properties either *in vivo* or *ex vivo* complicates their integration into finite element models, underscoring a significant limitation in current biomechanical assessments.

Furthermore, the interaction between the inorganic and organic components introduces non-linearities that are not easily captured by existing imaging techniques. For example, the role of collagen in resisting crack propagation and maintaining post-yield behavior is critical for bone toughness, but current imaging modalities cannot quantify the functional quality of collagen or its integration with the mineral phase (49, 77). Advances in techniques like Raman spectroscopy and nanoindentation have provided insights into these interactions in vitro, but these are not yet translatable to clinical imaging settings. Raman spectroscopy, including methods like surface-enhanced Raman scattering (SERS) and tip-enhanced Raman scattering (TERS), offers detailed molecular information and high spatial resolution (78). Nanoindentation, on the other hand, allows for precise measurement of mechanical properties at the nanoscale (79). Despite their potential, these techniques face challenges in clinical translation due to issues like signal interference and the complexity of in vivo environments.

Addressing the challenges of characterizing bone material properties from medical images requires significant advancements in imaging technologies and computational modeling. Techniques that integrate imaging with compositional analysis, such as dualenergy CT (DECT) or high-resolution peripheral QCT (HRpQCT), hold promise but remain in early stages of application (80). HR-pQCT is currently limited to extremities due to hardware and radiation constraints, making their use for larger regions like the proximal femur impractical. Empirical data from cadaveric studies could enhance finite element models, and future research could explore hybrid approaches combining high-resolution data with clinical imaging. Balancing radiation exposure with the need for detailed imaging is critical. Leveraging already-acquired clinical images for biomechanical modeling can reduce the need for additional scans. Standardizing imaging protocols in advance can further minimize radiation dose and costs while maintaining the necessary level of detail for accurate finite element analyses. Expanding our understanding of the material behavior of bone's components and improving the resolution and functionality of medical imaging will be critical for advancing finite element models and their clinical utility (81).

## 3.3.2 Anisotropy in bone mechanical properties and femoral strength

Anisotropy in bone mechanical properties refers to the variation in mechanical characteristics, such as Young's modulus and ultimate stress, depending on the orientation of the bone test sample, even when taken from the same site. Similarly, anisotropy in femoral strength indicates that the maximum force the femur can sustain before fracturing varies with the direction of the applied force. This anisotropy arises from bone's hierarchical structure and composition, including the alignment of collagen fibers, the distribution of hydroxyapatite crystals, and the trabecular architecture within the femoral head and neck (75, 82, 83). Cortical bone in the femoral shaft, for instance, is stiffer and stronger along the longitudinal axis, making it particularly effective at resisting axial loads during activities like walking and running (84). In contrast, the trabecular bone in the femoral head and neck features a highly complex, orientation-specific architecture designed to distribute stresses arising from multidirectional loading scenarios (85).

The majority of QCT-based finite element models employ simplified isotropic material assumptions for bone mechanical properties; however, they can still demonstrate the anisotropic nature of femoral strength due to the influence of bone geometry and heterogeneous material distribution. These models reveal that bone is more resistant to compression and tension in certain orientations while being more susceptible to shear forces in others (52, 86). Studies have shown that the femur's ability to withstand impact forces is highly dependent on the direction and magnitude of the force applied during a fall (24, 87, 88). For example, sideways falls, which are the most common fall scenario in elderly individuals, generate impact forces that are poorly aligned with the femur's primary axis of strength, significantly increasing the risk of fracture (24). Conversely, frontal or posterior falls may exert forces along directions that the femur is better adapted to withstand, reducing fracture risk (87, 88).

However, the isotropic models of bone mechanical properties inherently overlook the directional dependence of these properties, limiting their accuracy in simulating real-world loading conditions. To address this limitation, finite element models must incorporate anisotropic mechanical properties that reflect the true directional behavior of bone material. Achieving this level of precision requires advanced imaging and material characterization techniques, such as those capable of capturing collagen fiber orientation and mineral distribution, which are not yet widely accessible. This presents a significant barrier to advancing modeling accuracy and clinical applicability.

Current imaging modalities, such as QCT and DXA, are limited in their ability to comprehensively characterize bone composition, including inorganic minerals, organic proteins, and water, let alone provide detailed orientation-specific data on bone strength (89, 90). Incorporating composition- and microstructure-dependent mechanical properties and anisotropy into finite element models requires a deeper understanding of the hierarchical structure of bone, particularly the trabecular and cortical microstructures. Advanced imaging techniques, such as dual-energy computed tomography (DECT), offer promising avenues for distinguishing and quantifying bone components with greater specificity (91, 92). However, these techniques are still under development and face challenges such as resolution limitations and the accurate extraction of anisotropic mechanical properties. Overcoming these barriers is essential to achieving more precise and clinically relevant models for fracture risk assessment.

# 4 Image-based dynamics modeling of falls to predict impact forces

Falls are the leading cause of hip fractures, with over 95% of hip fractures attributed to falls from standing height (93). The forces generated during a fall frequently exceed the strength of the femur, resulting in fractures even in young, healthy individuals-let alone older adults, who often have compromised bone strength due to age-related changes or conditions like osteoporosis (94). However, only 2% of falls result in fractures (95, 96), highlighting the complex interplay between individual biomechanics, fall dynamics, and environmental factors. This low percentage underscores the importance of understanding how variables such as bone strength, fall-induced forces, body orientation during impact, and surface compliance collectively influence fracture outcomes. Fall experiments, even controlled fall testing, are neither ethical nor safe for elderly individuals. Image-based dynamics modeling offers a promising alternative for simulating falls and predicting impact forces by integrating subject-specific anatomical and biomechanical data derived from advanced medical imaging techniques.

This section explores the necessity and potential of subjectspecific dynamics modeling for fall simulations to predict impact forces. It discusses the importance of incorporating whole-body imaging data, such as DXA or QCT, to create accurate models, reviews the progress made in simulating falls from standing height, and examines the challenges that must be addressed to replicate real-world fall scenarios. By leveraging these advancements, the goal is to improve fracture risk prediction and develop more effective prevention strategies.

### 4.1 Subject-specific dynamics modeling of falls

Subject-specific factors, such as body height, weight, mass distribution, and flexibility, play a crucial role in determining the dynamics of a fall and the resulting impact forces (97). Generic models often fail to account for this variability, leading to inaccuracies in predicting impact forces and assessing fracture risks. For instance, a taller individual falling sideways may experience distinct dynamics and higher impact forces compared to a shorter individual under similar conditions. This variability underscores the need for personalized modeling. Subject-specific dynamics modeling of falls provides a more accurate approach by integrating individual characteristics, such as body dimensions, weight distribution, and flexibility, which significantly influence the trajectory and forces of a fall. Such precision is essential for reliably predicting impact forces and evaluating fracture risk.

Whole-body medical imaging techniques, such as DXA or QCT scans, offer valuable data on bone geometry, body composition, and soft tissue distribution, which are key parameters for developing subject-specific dynamics models of falls. For example, DXA scans provide detailed estimates of regional fat and muscle distribution, as illustrated in Figure 4, which directly influence body mass properties and thus the dynamics of a fall. These parameters play a critical role in determining how the body interacts with the ground during impact and how forces are absorbed and transmitted through various tissues. By integrating this personalized information, subject-specific models can more accurately simulate fall mechanics, enhancing the prediction of impact forces and the evaluation of fracture risks. Substantial progress has been made in the development of subject-specific dynamics models for simulating falls. For instance, Luo et al. (25) developed and validated a method for constructing subject-specific dynamics models using whole-body DXA images. These models demonstrated improved accuracy in predicting impact forces during sideways falls, showing better agreement with experimental data compared to traditional empirical functions (98). Similarly, Fleps et al. (99) introduced a dynamic inertia-driven sideways fall protocol that tested full cadaveric femur-pelvis constructs under realistic fall conditions. This approach aimed to enhance the prediction of impact loads and fracture risk by replicating the dynamics of real-world falls, thereby bridging the gap between laboratory testing and clinical relevance. Studies using finite element models combined with dynamics simulations have demonstrated the potential to predict impact forces and their distribution during falls. For instance, researchers have utilized whole-body musculoskeletal models derived from DXA and QCT data to simulate falls and calculate site-specific impact forces (26, 44, 100). Some of these models have been validated using experimental data, such as motion capture systems and force plates, providing evidence of their predictive accuracy (25, 101). Furthermore, machine learning approaches have been integrated into fall dynamics modeling to enhance the efficiency and accuracy of simulations. Algorithms trained on large datasets can optimize model parameters, such as fall orientation and joint motion, based on subject-specific input (102). These approaches have improved the ability to predict real-world fall scenarios and their associated forces (103).

Dynamic fall models estimate forces that are subsequently used as boundary and loading conditions in finite element simulations. While these forces provide critical inputs, simplifications—such as assuming uniform force distribution or neglecting soft tissue effects—may introduce translational losses. These approximations can affect the accuracy of fracture risk predictions. Future research should focus on improving the fidelity of force translation and accounting for individual-specific factors to enhance prediction reliability.

### 4.2 Challenges in simulating realworld falls

Simulating real-world falls presents significant challenges due to the inherent complexity and variability of fall dynamics and subjectspecific physiological factors (105, 106). Unlike controlled fall simulations, real-world falls are triggered by unpredictable and random events, such as tripping, slipping, or sudden loss of balance. These triggers introduce substantial variability in the initial conditions of the fall, including body posture, velocity, and the direction of movement at the onset of imbalance. Accurately replicating this randomness is crucial for realistic modeling but remains a significant hurdle.

Another critical factor is the reflexive response of muscles, which plays a pivotal role in influencing fall dynamics. When an individual loses balance, muscle reflexes are activated to counteract the fall, aiming to restore stability or reduce the severity of impact.



Muscle activation affects joint stiffness (107, 108), a key parameter in modulating the ability of the body to respond to destabilizing forces (109). Increased joint stiffness, resulting from heightened muscle activation, can stabilize the joints, preventing excessive movement that might exacerbate the fall. Conversely, insufficient muscle activation or weak muscle may lead to joint instability, increasing the likelihood of an uncontrolled descent. In addition to joint stiffness, muscle activation directly influences the actuator force generated by muscle fibers (110), which determines the strength and speed of corrective movements. For example, in a sideways fall, the hip abductor muscles play a crucial role in resisting lateral displacement of the torso (111), while the quadriceps and hamstrings stabilize the knees to reduce the impact force upon ground contact (112). These coordinated muscle activations help control body posture and orientation during the descent, potentially shifting the impact away from vulnerable areas like the hip.

The timing and intensity of muscle reflexes also vary between individuals, influenced by factors such as age, neuromuscular coordination, and physical fitness. Older adults, for instance, often exhibit delayed reflex responses and weaker muscle activation (113-115), which compromise their ability to mitigate the effects of a fall. In contrast, younger and physically active individuals tend to have faster and stronger reflexes, enhancing their capacity to absorb and dissipate impact energy. Additionally, muscle activation patterns influence the redistribution of body mass during a fall (97). For instance, active engagement of the arms and legs can alter the center of mass trajectory, reducing the likelihood of a high-impact collision at critical sites such as the hip. However, excessive or uncoordinated muscle activation can lead to counterproductive effects, such as increased rotational forces or misaligned body segments, potentially exacerbating the impact at the end of the fall (116).

Incorporating the randomness of fall triggers and the variability in muscle reflex responses into fall simulations requires sophisticated modeling approaches, along with subject-specific physiological and biomechanical parameters, which are extremely challenging to characterize. Current methodologies often rely on simplified assumptions regarding initial conditions and reflexive actions, limiting their ability to represent the full complexity of real-world falls. Advanced techniques, such as stochastic modeling to simulate random fall triggers and neuromuscular modeling to replicate reflexive muscle responses, are necessary to address these limitations. Overcoming these challenges is critical for improving the accuracy and applicability of fall dynamics models in assessing fracture risk and developing personalized prevention strategies.

## 5 Conclusion and future outlook

Recent advancements in image-based approaches for hip fracture risk assessment have significantly improved our understanding of the interplay between bone strength and fallinduced impact forces. Finite element (FE) models derived from imaging modalities such as DXA and QCT enable individualized assessments of femoral strength by capturing bone material properties, microstructure, and geometry. These models mark a notable improvement over traditional statistical tools by incorporating patient-specific risk factors. Similarly, subjectspecific dynamics modeling of falls has advanced the prediction of forces applied to the hip during real-world falls, offering the potential for more reliable fracture risk assessments. However, challenges remain in refining these approaches for improved accuracy and reliability.

Advancing biomechanical models for hip fracture risk assessment requires addressing several challenges:

- DXA-based FE models are limited by the projection of three-dimensional bone structures into two dimensions, which may introduce inaccuracies in estimating femoral strength. A significant limitation of DXA-based FE models is their sensitivity to body positioning during scanning, which can introduce variability in the estimated femoral strength and fracture risk. Ensuring consistent and accurate positioning is critical to improving the reliability of these models. QCT-based models offer greater anatomical detail but face barriers such as higher costs, increased radiation exposure, and limited accessibility. Both approaches require further improvements in accurately integrating material properties, such as bone density distribution and anisotropic strength, to enhance predictive accuracy.
- Characterizing bone mechanical properties based on medical images presents a significant challenge due to the difficulty of accurately mapping image-derived parameters, such as bone density, to mechanical properties like strength, stiffness, and toughness. Current methods often rely on empirical relationships that may not fully account for bone heterogeneity, anisotropy, and microstructural variations. To address these challenges, there is a need for more robust methodologies that couple advanced imaging techniques with experimental validation and multiscale modeling approaches, enabling more accurate prediction of mechanical behavior.
- In fall dynamics simulation, the complexity of real-world falls presents additional obstacles. Randomness in fall triggers, variability in fall trajectories, and reflexive muscle responses are difficult to replicate accurately. Current models often rely on simplified assumptions, limiting their ability to capture the variability observed in real-life scenarios. Advanced techniques, such as stochastic modeling for fall triggers and neuromuscular modeling for reflex responses, are needed to address these challenges and improve the reliability of impact force predictions.
- The integration of these image-based biomechanical models into clinical workflows remains limited due to technical and logistical constraints. Despite their detailed insights into hip fracture mechanisms, these models require further optimization for practical use in routine healthcare settings. Collaboration between engineers, clinicians, and imaging specialists is essential to bridge the gap between research and clinical practice.

In summary, image-based hip fracture risk assessment has made significant progress in offering patient-specific insights into fracture susceptibility. However, addressing technical challenges, refining modeling techniques, and facilitating clinical integration are critical for unlocking their full potential in improving fracture prevention and patient outcomes.

### Author contributions

YL: Conceptualization, Data curation, Formal Analysis, Funding acquisition, Investigation, Methodology, Project administration, Resources, Visualization, Writing – original draft, Writing – review & editing.

### Funding

The author(s) declare that financial support was received for the research, authorship, and/or publication of this article. The author's research work has been supported by the Natural Sciences and Engineering Research Council (NSERC) and the University of Manitoba, Canada.

### References

1. Feng J, Zhang C, Li B, Zhan S, Wang S, Song C. Global burden of hip fracture: The Global Burden of Disease Study. *Osteoporos Int.* (2024) 35:41–52. doi: 10.1007/s00198-023-06907-3

2. Collaborators G 2019 F. Global, regional, and national burden of bone fractures in 204 countries and territories, 1990-2019: a systematic analysis from the Global Burden of Disease Study 2019. *Lancet Healthy Longev.* (2021) 2:e580–92. doi: 10.1016/S2666-7568(21)00172-0

 Borgström F, Karlsson L, Ortsäter G, Norton N, Halbout P, Cooper C, et al. Fragility fractures in Europe: burden, management and opportunities. *Arch Osteoporos*. (2020) 15:59. doi: 10.1007/s11657-020-0706-y

4. Merlijn T, Swart KMA, van der Horst HE, Netelenbos JC, Elders PJM. Fracture prevention by screening for high fracture risk: a systematic review and meta-analysis. *Osteoporos Int.* (2020) 31:251–7. doi: 10.1007/s00198-019-05226-w

5. Engels A, Reber KC, Lindlbauer I, Rapp K, Büchele G, Klenk J, et al. Osteoporotic hip fracture prediction from risk factors available in administrative claims data – A machine learning approach. *PloS One.* (2020) 15:e0232969. doi: 10.1371/journal.pone.0232969

6. Sing C, Lin T, Bartholomew S, Bell JS, Bennett C, Beyene K, et al. Global epidemiology of hip fractures: secular trends in incidence rate, post-fracture treatment, and all-cause mortality. *J Bone Mineral Res.* (2023) 38:1064–75. doi: 10.1002/jbmr.4821

7. Cosman F, Lewiecki EM, Eastell R, Ebeling PR, Jan De Beur S, Langdahl B, et al. Goal-directed osteoporosis treatment: ASBMR/BHOF task force position statement 2024. J Bone Mineral Res. (2024) 39:1393–405. doi: 10.1093/jbmr/zjae119

8. Vilaca T, Schini M, Lui L-Y, Ewing SK, Thompson AR, Vittinghoff E, et al. The relationship between treatment-related changes in total hip BMD measured after 12, 18, and 24 mo and fracture risk reduction in osteoporosis clinical trials: the FNIH-ASBMR-SABRE project. *J Bone Mineral Res.* (2024) 39:1434–42. doi: 10.1093/jbmr/zjae126

9. Schwartz AV, Vittinghoff E, Bauer DC, Hillier TA, Strotmeyer ES, Ensrud KE, et al. Association of BMD and FRAX score with risk of fracture in older adults with type 2 diabetes. *JAMA*. (2011) 305:2184–92. doi: 10.1001/jama.2011.715

10. Johnell O, Kanis JA, Oden A, Johansson H, De Laet C, Delmas P, et al. Predictive value of BMD for hip and other fractures. *J Bone Mineral Res.* (2005) 20:1185–94. doi: 10.1359/JBMR.050304

11. Turcotte A-F, O'Connor S, Morin SN, Gibbs JC, Willie BM, Jean S, et al. Association between obesity and risk of fracture, bone mineral density and bone quality in adults: A systematic review and meta-analysis. *PloS One.* (2021) 16:e0252487. doi: 10.1371/journal.pone.0252487

### Conflict of interest

The author declares that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

### **Generative AI statement**

The author(s) declare that Generative AI was used in the creation of this manuscript. To select English words, to improve grammar and to search relevant journal articles.

### Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

12. Schini M, Johansson H, Harvey NC, Lorentzon M, Kanis JA, McCloskey EV. An overview of the use of the fracture risk assessment tool (FRAX) in osteoporosis. *J Endocrinol Invest.* (2023) 47:501–11. doi: 10.1007/s40618-023-02219-9

13. Richards C, Stevens R, Lix LM, McCloskey EV, Johansson H, Harvey NC, et al. Fracture prediction in rheumatoid arthritis: validation of FRAX with bone mineral density for incident major osteoporotic fractures. *Rheumatology*. (2025) 64:kead676. doi: 10.1093/rheumatology/kead676

14. Vandenput L, Johansson H, McCloskey EV, Liu E, Åkesson KE, Anderson FA, et al. Update of the fracture risk prediction tool FRAX: a systematic review of potential cohorts and analysis plan. *Osteoporos Int.* (2022) 33:2103–36. doi: 10.1007/s00198-022-06435-6

15. Mousa J, Peterson MN, Crowson CS, Achenbach SJ, Atkinson EJ, Amin S, et al. Validating the fracture risk assessment tool score in a US population-based study of patients with rheumatoid arthritis. *J Rheumatol.* (2023) 50:1279–86. doi: 10.3899/jrheum.2022-1293

16. Novel methods to evaluate fracture risk models | Journal of Bone and Mineral Research | Oxford Academic . Available online at: https://academic.oup.com/jbmr/article-abstract/26/8/1767/7597924?redirectedFrom=fulltext (Accessed October 23, 2024).

17. Marques A, Ferreira RJO, Santos E, Loza E, Carmona L, da Silva J. The accuracy of osteoporotic fracture risk prediction tools: a systematic review and meta-analysis. *Ann Rheum Dis.* (2015) 74:531. doi: 10.1136/annrheumdis-2015-eular.2984

18. Liu IT, Liang FW, Wang ST, Chang CM, Lu TH, Wu CH. The effects of falls on the prediction of osteoporotic fractures: epidemiological cohort study. *Arch Osteoporos*. (2021) 16:110. doi: 10.1007/s11657-021-00977-8

19. Harvey NC, Odén A, Orwoll E, Lapidus J, Kwok T, Karlsson MK, et al. Falls predict fractures independently of FRAX probability: A meta-analysis of the osteoporotic fractures in men (MrOS) study. *J Bone Mineral Res.* (2018) 33:510–6. doi: 10.1002/jbmr.3331

20. Beaudoin C, Moore L, Gagné M, Bessette L, Ste-Marie LG, Brown JP, et al. Performance of predictive tools to identify individuals at risk of non-traumatic fracture: a systematic review, meta-analysis, and meta-regression. *Osteoporos Int.* (2019) 30:721–40. doi: 10.1007/s00198-019-04919-6

21. Yosibash Z, Trabelsi N, Buchnik I, Myers KW, Salai M, Eshed I, et al. Hip fracture risk assessment in elderly and diabetic patients: combining autonomous finite element analysis and machine learning. *J Bone Mineral Res.* (2023) 38:876–86. doi: 10.1002/jbmr.4805

22. Galbusera F, Cina A, Panico M, Albano D, Messina C. Image-based biomechanical models of the musculoskeletal system. *Eur Radiol Exp.* (2020) 4:49. doi: 10.1186/s41747-020-00172-3

23. Boyd SK. Image-based finite element analysis. In: Hallgrimsson B, Sensen CW, editors. Advanced Imaging in Biology and Medicine. Berlin, Heidelberg, Germany: Springer (2009). p. 301–18.

24. Narsiri-Sarvi M, Luo Y. Sideways fall-induced impact force and its effect on hip fracture risk: a review. *Osteoporosis Int.* (2017) 28:2759–80. doi: 10.1007/s00198-017-4138-5

25. Luo Y, Sarvi MN, Sun P, Leslie W, Ouyang J. Prediction of impact force in sideways fall of the elderly by image-based subject-specific dynamics modeling. *Int Biomechanics.* (2014) 1:1–14. doi: 10.1080/23310472.2014.975745

26. Sarvi MN, Luo Y. A two-level subject-specific biomechanical model for improving prediction of hip fracture risk. *Clin Biomechanics*. (2015) 30:881–7. doi: 10.1016/j.clinbiomech.2015.05.013

27. Scott V, Wagar B, Sum A, Metcalfe S, Wagar L. A public health approach to fall prevention among older persons in Canada. *Clin Geriatr Med.* (2010) 26:705–18. doi: 10.1016/j.cger.2010.06.003

28. Yang Y, Komisar V, Shishov N, Lo B, Korall AM, Feldman F, et al. The effect of fall biomechanics on risk for hip fracture in older adults: A cohort study of video-captured falls in long-term care. *J Bone Miner Res.* (2020) 35:1914–22. doi: 10.1002/jbmr.4048

29. Parkkari J, Kannus P, Palvanen M, Natri A, Vainio J, Aho H, et al. Majority of hip fractures occur as a result of a fall and impact on the greater trochanter of the femur: a prospective controlled hip fracture study with 206 consecutive patients. *Calcif Tissue Int.* (2016) 65:183–7. doi: 10.1007/s002239900679

30. Lim S-K, Choi K, Heo NH, Kim Y, Lim J-Y. Characteristics of fragility hip fracture-related falls in the older adults: A systematic review. *J Nutrition Health Aging*. (2024) 28:100357. doi: 10.1016/j.jnha.2024.100357

31. Ferdous Z, Luo Y. Study of hip fracture risk by DXA-based patient-specific finite element model. *Bio-med Mater Eng.* (2015) 25:213–20. doi: 10.3233/BME-151271

32. Luo Y, Yang H. Assessment of hip fracture risk by cross-sectional strain-energy derived from DXA-based beam model. *Clin Biomechanics*. (2019) 63:48–53. doi: 10.1016/j.clinbiomech.2019.02.016

33. Luo Y. Empirical functions for conversion of femur areal and volumetric bone mineral density. *J Biomed Biol Eng.* (2019) 39:287–93. doi: 10.1007/s40846-018-0394-x

34. Luo Y, Ferdous Z, Leslie WD. A preliminary dual-energy X-ray absorptiometrybased finite element model for assessing osteoporotic hipfracture risk. *Proc Inst Mechanical Eng Part H: J Eng Med.* (2011) 225:1188–95. doi: 10.1177/0954411911424975

35. Luo Y, Ferdous Z, Leslie WD. Precision study of DXA-based patient-specific finite element modeling for assessing hip fracture risk. *Int J Numerical Methods Biomed Eng.* (2013) 29:615–29. doi: 10.1002/cnm.v29.5

36. Naylor KE, McCloskey EV, Eastell R, Yang L. Use of DXA-based finite element analysis of the proximal femur in a longitudinal study of hip fracture. *J Bone Miner Res.* (2013) 28:1014–21. doi: 10.1002/jbmr.1856

37. Yang L, Peel N, Clowes JA, McCloskey EV, Eastell R. Use of DXA-based structural engineering models of the proximal femur to discriminate hip fracture. *J Bone Miner Res.* (2009) 24:33–42. doi: 10.1359/jbmr.080906

 Terzini M, Aldieri A, Rinaudo L, Osella G, Audenino AL, Bignardi C. Improving the hip fracture risk prediction through 2D finite element models from DXA images: validation against 3D models. *Front Bioeng Biotechnol.* (2019) 7:220. doi: 10.3389/ fbioe.2019.00220

39. Danielson ME, Beck TJ, Karlamangla AS, Greendale GA, Atkinson EJ, Lian KY, et al. A comparison of DXA and CT based methods for estimating the strength of the femoral neck in post-menopausal women. *Osteoporos Int.* (2013) 24:1377–88. doi: 10.1007/s00198-012-2066-y

40. Luo Y, Ahmed S, Leslie WD. Automation of a DXA-based finite element tool for clinical assessment of hip fracture risk. *Comput Methods Programs Biomed.* (2018) 155:75–83. doi: 10.1016/j.cmpb.2017.11.020

41. Yang S, Leslie WD, Luo Y, Goertzen AL, Ahmed S, Ward LM, et al. Fullyautomated DXA-based finite element analysis stratifies hip fracture better than femoral neck bone mineral density: a cross-sectional study. *Osteoporosis Int.* (2018) 29:191–200. doi: 10.1007/s00198-017-4232-8

42. Dall'Ara E, Eastell R, Viceconti M, Pahr D, Yang L. Experimental validation of DXA-based finite element models for prediction of femoral strength. J Mechanical Behav Biomed Mater. (2016) 63:17–25. doi: 10.1016/j.jmbbm.2016.06.004

43. Yang L, Parimi N, Orwoll ES, Black DM, Schousboe JT, Eastell R. Association of incident hip fracture with the estimated femoral strength by finite element analysis of DXA scans in the Osteoporotic Fractures in Men (MrOS) study. *Osteoporos Int.* (2018) 29:643–51. doi: 10.1007/s00198-017-4319-2

44. Sarvi MN, Luo Y. Study of sex differences in the association between hip fracture risk and body parameters by DXA-based biomechanical modeling. *Bone*. (2016) 90:90–8. doi: 10.1016/j.bone.2016.06.006

45. Mochizuki T, Yano K, Ikari K, Kawakami K, Hiroshima R, Koenuma N, et al. Hip structure analysis by DXA of teriparatide treatment: A 24-month follow-up clinical study. *J Orthop.* (2016) 13:414–8. doi: 10.1016/j.jor.2016.09.001

46. Dudle A, Gugler Y, Pretterklieber M, Ferrari S, Lippuner K, Zysset P. 2D-3D reconstruction of the proximal femur from DXA scans: Evaluation of the 3D-Shaper software. *Front Bioeng Biotechnol.* (2023) 11:1111020. doi: 10.3389/fbioe.2023.1111020

47. Grassi L, Väänänen SP, Jehpsson L, Ljunggren Ö, Rosengren BE, Karlsson MK, et al. 3D finite element models reconstructed from 2D dual-energy X-ray

absorptiometry (DXA) images improve hip fracture prediction compared to areal BMD in osteoporotic fractures in men (MrOS) Sweden cohort. J Bone Miner Res. (2023) 38:1258–67. doi: 10.1002/jbmr.4878

48. Saito M, Marumo K. Collagen cross-links as a determinant of bone quality: a possible explanation for bone fragility in aging, osteoporosis, and diabetes mellitus. *Osteoporos Int.* (2010) 21:195–214. doi: 10.1007/s00198-009-1066-z

49. Viguet-Carrin S, Garnero P, Delmas PD. The role of collagen in bone strength. Osteoporos Int. (2006) 17:319-36. doi: 10.1007/s00198-005-2035-9

50. Banks KP, Farrell MB, Gunther RS, McWhorter NE, Byerly DW, Peacock JG. Improving DXA quality by avoiding common technical and diagnostic pitfalls: part 1. *J Nucl Med Technol.* (2023) 51:167–75. doi: 10.2967/jnmt.122.264885

51. Knowles NK, Reeves JM, Ferreira LM. Quantitative Computed Tomography (QCT) derived Bone Mineral Density (BMD) in finite element studies: a review of the literature. J Exp Orthopaedics. (2016) 3:1–16. doi: 10.1186/s40634-016-0072-2

52. Kheirollahi H, Luo Y. Understanding hip fracture by QCT-based finite element modeling. J Med Biol Eng. (2017) 37:686-94. doi: 10.1007/s40846-017-0266-9

53. Carpenter RD. Finite element analysis of the hip and spine based on quantitative computed tomography. *Curr Osteoporos Rep.* (2013) 11:156–62. doi: 10.1007/s11914-013-0141-8

54. Zysset P, Dall'Ara E, Varga P, Pahr DH. Finite element analysis for prediction of bone strength. *BoneKEy Rep.* (2013) 2:1–9. doi: 10.1038/bonekey.2013.120

55. Faisal TR, Luo Y. Study of fracture risk difference in left and right femur by QCTbased FEA. *Biomed Eng Online*. (2017) 16:116. doi: 10.1186/s12938-017-0407-y

56. Dragomir-Daescu D, Op Den Buijs J, McEligot S, Dai YF, Entwistle RC, Salas C, et al. Robust QCT/FEA models of proximal femur stiffness and fracture load during a sideways fall on the hip. *Ann Biomed Eng.* (2011) 39:742–55. doi: 10.1007/s10439-010-0196-y

57. Mirzaei M, Keshavarzian M, Naeini V. Analysis of strength and failure pattern of human proximal femur using quantitative computed tomography (QCT)-based finite element method. *Bone.* (2014) 64:108–14. doi: 10.1016/j.bone.2014.04.007

58. Dall'Ara E, Luisier B, Schmidt R, Kainberger F, Zysset P, Pahr D. A nonlinear QCT-based finite element model validation study for the human femur tested in two configurations *in vitro*. *Bone*. (2013) 52:27–38. doi: 10.1016/j.bone.2012.09.006

59. Shen J, Nielson CM, Marshall LM, Lee DC, Keaveny TM, Orwoll ES. The association between BMI and QCT-derived proximal hip structure and strength in older men: a cross-sectional study. *J Bone Miner Res.* (2015) 30:1301–8. doi: 10.1002/jbmr.2450

60. Black DM, Bouxsein ML, Marshall LM, Cummings SR, Lang TF, Cauley JA, et al. Proximal femoral structure and the prediction of hip fracture in men: a large prospective study using QCT. *J Bone Miner Res.* (2008) 23:1326–33. doi: 10.1359/jbmr.080316

61. Kheirollahi H, Luo Y. Assessment of hip fracture risk using cross-section strain energy determined from QCT-based finite element model. *BioMed Res Int.* (2015) 2015:Article ID 413839. doi: 10.1155/2015/413839

62. Carpenter RD, Beaupr? GS, Lang TF, Orwoll ES, Carter DR. New QCT analysis approach shows the importance of fall orientation on femoral neck strength. *J Bone Miner Res.* (2005) 20:1533–42. doi: 10.1359/JBMR.050510

63. Christiansen BA, Kopperdahl DL, Kiel DP, Keaveny TM, Bouxsein ML. Mechanical contributions of the cortical and trabecular compartments contribute to differences in age-related changes in vertebral body strength in men and women assessed by QCT-based finite element analysis. *J Bone Mineral Res.* (2011) 26:974–83. doi: 10.1002/jbmr.287

64. Engelke K, Fuerst T, Dasic G. al. et. Regional distribution of spine and hip QCT BMD responses after one year of once-monthly ibandronate in postmenopausal osteoporosis. *Bone.* (2010) 46:1626–32. doi: 10.1016/j.bone.2010.03.003

65. Engelke K, Lang T, Khosla S, Qin L, Zysset P, Leslie WD, et al. Clinical use of quantitative computed tomography (QCT) of the hip in the management of osteoporosis in adults: the 2015 ISCD official positions - part I. *J Clin Densitomet.* (2015) 18:338–58. doi: 10.1016/j.jocd.2015.06.012

66. Crawford RP, Cann CE, Keaveny TM. Finite element models predict *in vitro* vertebral body compressive strength better than quantitative computed tomography. *Bone.* (2003) 33:744–50. doi: 10.1016/S8756-3282(03)00210-2

67. Boskey AL. Bone composition: relationship to bone fragility and antiosteoporotic drug effects. *BoneKey Rep.* (2013) 447. doi: 10.1038/bonekey.2013.181

68. Boskey AL, Wright TM, Blank RD. Collagen and bone strength. J Bone Mineral Res. (1999) 14:330–5. doi: 10.1359/jbmr.1999.14.3.330

69. Nyman JS, Roy A, Acuna RL, Shen X, Tyler JH, Wang X. The influence of water removal on the strength and toughness of cortical bone. *J Biomech*. (2006) 39:931–8. doi: 10.1016/j.jbiomech.2005.01.012

70. Granke M, Does MD, Nymna JS. The role of water compartments in the material properties of cortical bone. *Calcif Tissue Int.* (2015) 97:292–307. doi: 10.1007/s00223-015-9977-5

71. Lang TF. Quantitative computed tomography. Radiol Clin North Am. (2010) 48:589-600. doi: 10.1016/j.rcl.2010.03.001

72. Adams JE. Advances in bone imaging for osteoporosis. *Nat Rev Endocrinol.* (2013) 9:28-42. doi: 10.1038/nrendo.2012.217

73. SChileo E, Taddei F, Malandrino A, Cristofolini L, Viceconti M. Subject-specific finite element models can accurately predict strain levels in long bones. *J Biomech*. (2007) 40:2982–9. doi: 10.1016/j.jbiomech.2007.02.010

74. Rodriguez-Palomo A, Østergaard M, Birkedal H. Bone hierarchical structure: heterogeneity and uniformity. *Adv Funct Mater.* (2024) 34:2307026. doi: 10.1002/adfm.202307026

75. Fratzl P, Weinkamer R. Nature's hierarchical materials. *Prog Mater Sci.* (2007) 52:1263–334. doi: 10.1016/j.pmatsci.2007.06.001

76. Boskey AL, Coleman R. Aging and bone. J Dental Res. (2010) 89:1333-48. doi: 10.1177/0022034510377791

77. Burr DB. The contribution of the organic matrix to bone's material properties. Bone. (2002) 31:8-11. doi: 10.1016/S8756-3282(02)00815-3

78. Buckley K, Kerns JG, Vinton J, Gikas PD, Smith C, Parker AW, et al. Towards the *in vivo* prediction of fragility fractures with Raman spectroscopy. *J Raman Spectrosc.* (2015) 46:610–8. doi: 10.1002/jrs.v46.7

79. Wang C-F, O'Callahan BT, Krayev A, El-Khoury PZ. Nanoindentationenhanced tip-enhanced Raman spectroscopy. *J Chem Phys.* (2021) 154:241101. doi: 10.1063/5.0056541

80. de Bakker CMJ, Tseng W-J, Li Y, Zhao H, Liu XS. Clinical evaluation of bone strength and fracture risk. *Curr Osteoporos Rep.* (2017) 15:32–42. doi: 10.1007/s11914-017-0346-3

81. Williams S, Khan L, Licata AA. DXA and clinical challenges of fracture risk assessment in primary care. CCJM. (2021) 88:615–22. doi: 10.3949/ccjm.88a.20199

82. Reznikov N, Bilton M, Lari L, Stevens MM, Kröger R. Fractal-like hierarchical organization of bone begins at the nanoscale. *Science*. (2018) 360:eaao2189. doi: 10.1126/science.aao2189

83. Tzaphlidou M. Bone architecture: collagen structure and calcium/phosphorus maps. J Biol Phys. (2008) 34:39–49. doi: 10.1007/s10867-008-9115-y

84. Cuppone M, Seedhom BB, Berry E, Ostell AE. The longitudinal young's modulus of cortical bone in the midshaft of human femur and its correlation with CT scanning data. *Calcif Tissue Int*. (2004) 74:302–9. doi: 10.1007/s00223-002-2123-1

85. Cauley JA, Blackwell T, Zmuda JM, Fullman RL, Ensrud KE, Stone KL, et al. Correlates of trabecular and cortical volumetric bone mineral density (vBMD) at the femoral neck and lumbar spine: The osteoporotic fractures in men study (MrOS). *J Bone Miner Res.* (2010) 25:1958–71. doi: 10.1002/jbmr.86

86. Trabelsi N, Yosibash Z. Patient-specific finite-element analyses of the proximal femur with orthotropic material properties validated by experiments. *J Biomechanical Eng.* (2011) 133. doi: 10.1115/1.4004180

87. Ford CM, Keaveny TM, Hayes WC. The effect of impact direction on the structural capacity of the proximal femur during falls. *J Bone Mineral Res.* (1996) 11:377–83. doi: 10.1002/jbmr.5650110311

88. Pretty SP, Levine IC, Laing AC. Anatomically aligned loading during falls: influence of fall protocol, sex and trochanteric soft tissue thickness. *Ann BioMed Eng.* (2021) 49:3267–79. doi: 10.1007/s10439-021-02852-6

89. Choksi P, Jepsen KJ, Clines GA. The challenges of diagnosing osteoporosis and the limitations of currently available tools. *Clin Diabetes Endocrinol.* (2018) 4:12. doi: 10.1186/s40842-018-0062-7

90. Wang F, Zheng L, Theopold J, Schleifenbaum S, Heyde C-E, Osterhoff G. Methods for bone quality assessment in human bone tissue: a systematic review. *J Orthopaedic Surg Res.* (2022) 17:174. doi: 10.1186/s13018-022-03041-4

91. Engelke K, Libanati C, Fuerst T, Zysset P, Genant HK. Advanced CT based *in vivo* methods for the assessment of bone density, structure, and strength. *Curr Osteoporos Rep.* (2013) 11:246–55. doi: 10.1007/s11914-013-0147-2

92. Koch V, Hokamp NG, Albrecht MH, Gruenewald LD, Yel I, Borggrefe J, et al. Accuracy and precision of volumetric bone mineral density assessment using dualsource dual-energy versus quantitative CT: a phantom study. *Eur Radiol Exp.* (2021) 5:43. doi: 10.1186/s41747-021-00241-1

93. Barceló M, Casademont J, Mascaró J, Gich I, Torres OH. Indoor falls and number of previous falls are independent risk factors for long-term mortality after a hip fracture. *Aging Clin Exp Res.* (2023) 35:2483–90. doi: 10.1007/s40520-023-02551-3

94. Komisar V, Robinovitch SN. The role of fall biomechanics in the cause and prevention of bone fractures in older adults. *Curr Osteoporos Rep.* (2021) 19:381–90. doi: 10.1007/s11914-021-00685-9

95. Ardaneh M, Fararouei M, Hassanzadeh J. Factors contributing to falls leading to fracture among older adults. *Popul Ageing*. (2023) 16:121–35. doi: 10.1007/s12062-021-09326-6

96. Leavy B, Byberg L, Michaëlsson K, Melhus H, Åberg AC. The fall descriptions and health characteristics of older adults with hip fracture: a mixed methods study. *BMC Geriatrics.* (2015) 15:40. doi: 10.1186/s12877-015-0036-x

97. Pretty SP, Martel DR, Laing AC. The influence of body mass index, sex, & Muscle activation on pressure distribution during lateral falls on the hip. *Ann BioMed Eng.* (2017) 45:2775–83. doi: 10.1007/s10439-017-1928-z

98. Nasiri M, Luo Y, Sun P, Ouyang J. Experimental validation of subject-specific dynamics model for predicting impact force in sideways fall. *J Biomed Sci Eng.* (2014) 7:405–18. doi: 10.4236/jbise.2014.77043

99. Fleps I, Vuille M, Melnyk A, Ferguson SJ, Guy P, Helgason B, et al. A novel sideways fall simulator to study hip fractures ex vivo. *PloS One.* (2018) 13:e0201096. doi: 10.1371/journal.pone.0201096

100. Fleps I, Guy P, Ferguson SJ, Cripton PA, Helgason B. Explicit finite element models accurately predict subject-specific and velocity-dependent kinetics of sideways fall impact. J Bone Mineral Res. (2019) 34:1837–50. doi: 10.1002/jbmr.3804

101. Bliven EK, Fung A, Baker A, Fleps I, Ferguson SJ, Guy P, et al. How accurately do finite element models predict the fall impact response of ex vivo specimens augmented by prophylactic intramedullary nailing? *J Orthopaedic Res.* (2025) 43:396-406. doi: 10.1002/jor.25984

102. Tufisi C, Praisach Z-I, Gillich G-R, Bichescu AI, Heler T-L. Forward fall detection using inertial data and machine learning. *Appl Sci.* (2024) 14:10552. doi: 10.3390/app142210552

103. Lim ZK, Connie T, Goh MKO, Saedon N 'Izzati B. Fall risk prediction using temporal gait features and machine learning approaches. *Front Artif Intell.* (2024) 7:1425713. doi: 10.3389/frai.2024.1425713

104. Luo Y. Image-based multilevel biomechanical modeling for fall-induced hip fracture. New York, USA: Springer (2017).

105. Broadley RW, Klenk J, Thies SB, Kenney LPJ, Granat MH. Methods for the real-world evaluation of fall detection technology: A scoping review. *Sensors*. (2018) 18:2060. doi: 10.3390/s18072060

106. Tchalla AE, Dufour AB, Travison TG, Habtemariam D, Iloputaife I, Manor B, et al. Patterns, predictors, and outcomes of falls trajectories in older adults: the MOBILIZE boston study with 5 years of follow-up. *PloS One.* (2014) 9:e106363. doi: 10.1371/journal.pone.0106363

107. Jeong S, Nishikawa K. The force response of muscles to activation and length perturbations depends on length history. *J Exp Biol.* (2023) 226:jeb243991. doi: 10.1242/jeb.243991

108. Seth A, Hicks JL, Uchida TK, Habib A, Dembia CL, Dunne JJ, et al. OpenSim: Simulating musculoskeletal dynamics and neuromuscular control to study human and animal movement. *PloS Comput Biol.* (2018) 14:e1006223. doi: 10.1371/journal.pcbi.1006223

109. Krause A, Freyler K, Gollhofer A, Stocker T, Brüderlin U, Colin R, et al. Neuromuscular and kinematic adaptation in response to reactive balance training – a randomized controlled study regarding fall prevention. *Front Physiol.* (2018) 9:1075. doi: 10.3389/fphys.2018.01075

110. Wang Y, Liu C, Ren L, Ren L. Load-dependent variable gearing mechanism of muscle-like soft actuator. J Bionic Eng. (2022) 19:29–43. doi: 10.1007/s42235-021-00129-1

111. Lanza MB, Rock K, Marchese V, Addison O, Gray VL. Hip abductor and adductor rate of torque development and muscle activation, but not muscle size, are associated with functional performance. *Front Physiol.* (2021) 12:744153. doi: 10.3389/fphys.2021.744153

112. Walsh M, Boling MC, McGrath M, Blackburn JT, Padua DA. Lower extremity muscle activation and knee flexion during a jump-landing task. J Athletic Training. (2012) 47:406–13. doi: 10.4085/1062-6050-47.4.17

113. Clark BC. Neural mechanisms of age-related loss of muscle performance and physical function. J Gerontol: Ser A. (2023) 78:8–13. doi: 10.1093/gerona/glad029

114. Jeon W, Ramadan A, Whitall J, Alissa N, Westlake K. Age-related differences in lower limb muscle activation patterns and balance control strategies while walking over a compliant surface. *Sci Rep.* (2023) 13:16555. doi: 10.1038/s41598-023-43728-0

115. Cano Porras D, Jacobs JV, Inzelberg R, Bahat Y, Zeilig G, Plotnik M. Patterns of whole-body muscle activations following vertical perturbations during standing and walking. *J NeuroEng Rehabil.* (2021) 18:75. doi: 10.1186/s12984-021-00836-0

116. Adams M, Gordt-Oesterwind K, Bongartz M, Zimmermann S, Seide S, Braun V, et al. Effects of physical activity interventions on strength, balance and falls in middle-aged adults: A systematic review and meta-analysis. *Sports Med - Open.* (2023) 9:61. doi: 10.1186/s40798-023-00606-3